Author’s response to reviews

Title: Neuromyelitis Optica Spectrum Disorder with Increased Aquaporin 4 Microparticles Prior to Autoantibodies in Cerebrospinal Fluid: A Case Report of a PANDAS patient

Authors:

Susanne Bejerot (susanne.bejerot@gmail.com)
Eva Hesselmark (eva.hesselmark@ki.se)
Fariborz Mobarez (fariborz.mobarez@ki.se)
Håkan Wallén (hakan.wallen@ds.se)
Max Hietala (albert.hietala@ki.se)
Rolf Nybom (rolf.nybom@ki.se)
Lennart Wetterberg (lennart.wetterberg@ki.se)

Version: 1 Date: 25 Oct 2018

Author’s response to reviews:

Dear Dr. Moscote-Salazar,

Thank you for your positive review of our manuscript. We have not found any suggestions for editing or clarification from the reviewers or from the editorial office, and this we resubmit the manuscript unchanged. We have made a formatting change to the reference list, and hope that the references can now be validated. Please let me know if the manuscript needs any more editing before publication.

Thank you again for considering our manuscript for publication.

Kind regards,

Lennart Wetterberg and the authors.